Author’s response to reviews

Title: LMA Gastro™ Airway is feasible during upper gastrointestinal interventional endoscopic procedures in high risk patients: a single-center observational study

Authors:

Axel Schmutz (axel.schmutz@uniklinik-freiburg.de)

Thomas Loeffler (thomas.loeffler@uniklinik-freiburg.de)

Arthur Schmidt (arthur.schmidt@uniklinik-freiburg.de)

Ulrich Goebel (ulrich.goebel@uniklinik-freiburg.de)

Version: 3 Date: 09 Jan 2020

Author’s response to reviews:

Dear Editor,

Editor Comments:

1. Consent for publication

Please amend your “Consent for publication” section in your Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.

Authors comments: Done as requested,

Thank you!